Great Basin Incident Commander Conference Call Agenda 2016

Date: Friday, August 12, 2016

| Note taker: Gina Bald

GB MAC/GACC Coordinator:

Agenda/Incident Commander Discussion Points: Summarize the following in 4 min or less:

. Fire Size / % contained / # of people / Cost to date

. Significant changes since the last 209

. BRIEF fire synopsis /Critical Incident Information (use Directional vs DIVS’s)
. Political Issues / Community Meetings

o Closures / Evacuations / Public Safety Issues

. Major Safety Concerns / Injuries / Accidents

. Critical Resource Needs

. Resources Available to Reassign/Share/Demob

Predictive Services:
Weather update:
Shelby Law

High pressure over the GB — warming temps and light winds. Sunday:

Central ID mtns and W. WY — thunderstorms. Mid-week: return of

monsoonal moisture.

GBCC Update:
Gina Dingman

Light IA. 8 new for 27 acres

Large Fire Growth: 3000

Reassigning resources as they come available. T1 Crews scarce.
Competition for resources is substantial.

In the GB: 2 MAFFs, 2 Scoopers, 5 AT (1 D/O), 1 VLAT, 21 SEATS
(2D/0, 1 0/S), 1 ASM (avail at 1630), 5 LP (1 avail at 1630)
Helos Committed: 16 T1,157T2,25T3

MAC / GBCC
Coordinator
Derrek Hartman

Nothing to Report

1

Fire: Pioneer

IC: Wilkins, South
Zone, Chadwick,
North Zone
Report: Morales,
Chadwick

65285 / 37% /1459 /34.7M

Increase of 1600 acres

South Zone: DIV B: Main activity. Burnout and moving fire. DIV
GG: Finished contingency line. Firing into existing burn scar.
North Zone: Thanks for helping with transition plan. GB Team 6
handling North of the South Fork of the Payette River. Critical
burnout today. East of North Zone: Lowman and Deadwood
drainage — nasty ground. Sunday: winds hot/dry. Fear that fire
will take off and run north to Deadwood drainage. A lot of debris
coming down on Hwy 17.

Critical Resource Needs: 3 T1 Crews, 3 T2IA for support on
burnout and resources timing out

Closures: Forest, Hwy 17 closed. Hwy 21 open during the day,
closed at night.




e Community meeting: Sunday in Lowman
e Injuries: None
e No Resources available to reassign or release

2
Fire: Strawberry
IC: Kuo

4594 /30% /365 / 1.2M

e Increase of 100 acres

e Suppression strategy for values at risk

e North: Good progress. Protecting powerlines.

e East and West: Direct attack. Flatter, sage brush —good progress.

e South: Heavier timber, bug kill. Helos and retardant, additional
dip sites. Structure group for the Park. Bonneville Cutthroat
Trout — relocation of survivors.

e Park has given IA responsibility

e Political Issues: Joint delegation between BLM and White Pine
County

e Community Meeting in Baker tonight

e Closures: Portion of Park, upper campgrounds. Cave and Hwy 50
still open.

e No Critical Resource Needs. T2 Helo, 2 Hand crews

e No Resources available for release/reassignment

3 1800/ 10% / 215 / $350000
Fire: Rough e Heavy use of fixed and rotor aircraft
IC: Shanahan e Two periods of significant spotting and slopover
e WFM scouting North end
e Need a couple of good days of work in Rough Creek with current
resources.
e Critical Resource Needs: None
e Resources available for release/reassignment: None
e Injuries: None
e Safety Concerns: Snags, rough terrain
4 7695/ 65% / 312 / $430000

Fire: Overland
IC: Cain

e Decrease in acreage due to mapping

e East and Southeast Flanks: Fair amount of heat — finding spots. 1
—1 % mile of unchecked line. Steep.

e No Political Issues

e No Injuries

e No Critical Resource Needs. Maintain Hotshots and T3 Helo

e No Resources available for release/reassignment. Maintain
current aviation resources.

5
Fire: Broad Mouth
IC: Ames

20616 /91% /71 / 1.896M
e Estimated containment August 13
e Maintaining current aviation resources

No Report

6 788 /90% /93 /1.2M

Fire: Dry Creek e Estimated containment this evening at 2000

IC: McCoy e Reassigned 3 IHCs to Pioneer this morning — enroute

e 1IHC loaned to Vienna — available when they return




e Transition to T4 Saturday 0600
e Injuries: 1 ankle. Transported to camp and swapped out

7 3100/ 43% / 113 / $370000
Fire: Powerline e Hope for 100% containment this evening
IC: Williams e Political Issues: Tribal sensitive cultural areas
e No Injuries
e No Critical Resource Needs
e Resources available for release/reassignment: Local resources,
dozers
8 2642 /15% /72 / 1.257M
Fire: Box Canyon e Updated map
IC: Pilgrim e Added Long Term Strategy Analysis
e Creeping, isolated areas
e Received heavy precipitation in higher elevations
e Closures: Opened Ledge Fork Campground. Trails, dirt roads
remain closed.
e Injuries: None
e Critical Resource Needs: None
e Resources to assign: None
e Transition on Saturday 0600 to new T3 organization. In-Briefing
today.
10 400 / 40% / 17 / $45000
Fire: Grimms e Active, doubled in size
IC: Woolsey e Two Flanks: South and West — doing good, holding
Report: Peel e Within strategy and box

e Expecting favorable weather

e No Critical Resource Needs

e No Resources available to reassign

e Expecting 600 acres when said and done

Date/Time of Next Call:

8/13 @ 0800 MT / 0900 PT




